ORIGINAL RESEARCH

Testing a Cloud-Based Model for Active Surveillance of Medical Devices with Analyses of Coronary Stent Safety Using the Data Extraction and Longitudinal Trend Analysis (DELTA) System

Joseph P Drozda Jr¹^{*}, Henry Ssemaganda^{2,*}, Edward A Frankenberger^{3,*}, Eric Brandt^{4,*}, Susan Robbins^{2,*}, Neha Khairnar^{2,*}, Alexandra Cha^{5,*}, Frederic S Resnic^{6,*}

¹Retiree, Mercy, Chesterfield, MO, USA; ²Comparative Effectiveness Research Institute, Lahey Hospital and Medical Center, Burlington, MA, USA; ³Freenome, South San Francisco, CA, USA; ⁴Research, Mercy, Chesterfield, MO, USA; ⁵Booz Allen Hamilton, Bethesda, MD, USA; ⁶Division of Cardiovascular Medicine, Lahey Hospital and Medical Center, Burlington, MA, USA

*These authors contributed equally to this work

Correspondence: Joseph P Drozda Jr, Tel +1 314 308 1732, Email jpdrozda@charter.net

Objective: To demonstrate the use of the Data Extraction and Longitudinal Trend Analysis (DELTA) system in the National Evaluation System for health Technology's (NEST) medical device surveillance cloud environment by analyzing coronary stent safety using real world clinical data and comparing results to clinical trial findings.

Design and Setting: Electronic health record (EHR) data from two health systems, the Social Security Death Master File, and device databases were ingested into the NEST cloud, and safety analyses of two stents were performed using DELTA.

Participants and Interventions: This is an observational study of patients receiving zotarolimus drug-eluting coronary stents (ZES) or everolimus eluting coronary stents (EES) between July 1, 2015 and December 31, 2017.

Results: After exclusions, 3334 patients receiving EES and 1002 receiving ZES were available for study. Analysis using inverse probability weighting showed no significant difference in one-year mortality or major adverse cardiac events (MACE) for EES compared to ZES [Mortality Odds Ratio 0.94 (95% CI 0.81–1.175); p = 0.780] [MACE Odds Ratio 1.04 (95% CI 0.92–1.16; p = 0.551]). Analysis using propensity matching showed no significant difference in EES one-year mortality (547 of 992 alive and available after censoring) compared to ZES (546 of 992) [Log-Rank statistic 0.3348 (p = 0.563)].

Conclusion: Automated cloud-based medical device safety surveillance using EHR data is feasible and was efficiently performed using DELTA. No statistically significant differences in 1-year safety outcomes between ZES and EES were identified using two statistical approaches, consistent with randomized trial findings.

Plain Language Summary:

- What is already known on this topic-The National Evaluation System for health Technology (NEST) Coordinating Center has developed a prototype cloud-based medical device surveillance system designed to capture clinical data obtained during routine care along with automated analysis tools to monitor device safety and performance in a real-world setting. DELTA is a suite of open-source active surveillance software tools that has been successfully deployed in other data environments and was being evaluated in the NEST cloud.
- What this study adds-This paper reports on the validation of DELTA methods deployed in the NEST cloud environment (appendix) as well as the initial demonstration of active safety surveillance for two commonly used coronary stent devices. Electronic health record data on the two coronary stents from two health systems were ingested into the Cloud, linked to the Social Security Death Master File along with a reference device database, and analyzed with DELTA as a prototype of an active safety surveillance system.
- How this study might affect research, practice, or policy-Consistent with clinical trials and a prior independent analysis, no statistically significant differences in clinical outcomes were found between the two stents. Importantly, the results support the feasibility of using the NEST multi-health system cloud for monitoring medical device safety and effectiveness. The system could

© 2024 Drozda Jr et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/ the work you hereby accept the Terms.Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. for permission for commercial uses of this work, please see paragraphs 4.2 and 5 of our Terms (https://www.dovepress.com/terms.php). minimize the risks associated with late recognition of device safety issues and support assessments of medical device value in the real world.

Keywords: active surveillance, safety-based medical device withdrawals, real-world evidence, cardiac devices, outcomes assessment (health care)

Introduction

Timely identification of medical device safety concerns, particularly early safety signal identification, has been a challenge using current approaches, eg, post-approval studies that are slow to enroll and often not completed and "passive" surveillance through adverse event reporting. The National Evaluation System for health Technology Coordinating Center (NESTcc) sought to strengthen the system of post-market medical device safety and effectiveness evaluation using real-world data (RWD).¹ The NESTcc explored a cloud-based infrastructure for the primary purpose of active surveillance using RWD from electronic health records (EHRs) and other sources.² The Data Extraction and Longitudinal Trend Analysis (DELTA) system was chosen as an active surveillance tool for identifying safety signals in NESTcc's analytics environment. DELTA is an open-source software package that supports a variety of statistical surveillance methodologies that can be configured to ingest new data and perform sequential analyses on predefined schedules. [See Supplemental Content.]

DELTA accesses databases containing patient characteristics, treatment exposures, and outcomes, and then utilizes a variety of statistical methodologies, including risk-adjusted prospective surveillance analyses, survival studies, sequential analyses, and extensive propensity matching algorithms, to analyze the data. The application provides a single graphical user interface to configure the analyses which are executed via scripts written in the R programming language.

The full spectrum of DELTA analytic methods was initially tested and validated within the NEST cloud environment using complex synthetic datasets and historical clinical data from a single healthcare system. The validation analysis using complex synthetic datasets demonstrated comprehensive accuracy of the DELTA cloud implementation as compared to results generated on a single, local, server (<u>Supplemental Content Tables S1</u> and <u>S2</u>). Separately, the DELTA cloud implementation was used to evaluate the safety of drug-eluting stents as compared with bare-metal stents, using previously analyzed data, and accurately replicated the published analysis (Supplemental Content Table S3 and Figure S1).

The purpose of the current study was to further demonstrate the feasibility of using DELTA for cloud-based active surveillance by analyzing the relative clinical performance of two commonly used drug eluting coronary stents [zotarolimus-eluting stents (ZES) and everolimus-eluting stents (EES)] using RWD from two health systems as a prototype for future cloud-based active surveillance efforts. Importantly, the selection of ZES and EES as devices to evaluate through this validation exercise was based on frequency of device usage and structured data availability, given the harmonization of data element definitions between Mercy and Lahey through their common use of the National Cardiovascular Data Registry's (NCDR[®]) CathPCI Registry[®].

Methods

98

Data Sources

Data were sourced from both the Mercy Building Unique Device Identifiers (UDIs) Into Longitudinal Data for Medical Device Evaluation (BUILD) database and the Lahey Health Observational Database Environment (LHODE). BUILD is an FDA and industry-funded initiative in which three large health systems created UDI enriched research databases (UDIRs) and formed a distributed data network.³ The only BUILD data used in the current study were from the Mercy UDIR, which has been described in detail elsewhere⁴ and which was extensively redesigned to conform to the BUILD common data model (CDM). The BUILD CDM is comprised of elements from the CDM developed for the FDA-sponsored Sentinel program that include longitudinal electronic health record (EHR) and payer data⁵ as well as cardiology-specific tables that closely followed CathPCI data content.⁶ In addition, the CDM included tables of publicly accessible device data from the FDA Global UDI Database (GUDID) available from AccessGUDID at the National Library of Medicine⁷ and from the "augmented UDI" (AUDI) table containing nine stent-specific attributes selected by an expert panel of cardiologists during the Mercy

demonstration project.⁸ In addition to the device data, Mercy UDIR data consist of all discrete fields within the EHR including billing, data from Mercy's cardiac cath lab clinical systems, and data from systems used to report to the CathPCI Registry[®]. Likewise, the Lahey Health submitted data also used all EHR-derived data, including billing information, as well as the CathPCI abstracted data submissions.

The Lahey Health Comparative Effectiveness Research Institute (CERI) maintains a data warehouse that is dynamically updated with structured demographic, clinical, laboratory, and billed claims data for all patient encounters in any of the Lahey Health System's facilities, extracted from the Lahey electronic health data system. Additionally, data elements from Lahey's CathPCI submission database are linked to CERI data warehouse to provide additional procedural information, patient characteristics, and device-specific information related to the use of coronary stents. The Lahey CERI data warehouse is also linked to the Social Security Death Master File for ascertainment of vital status at 1-year post-procedure.

Study Design

All patients receiving ZES or EES within the Mercy Health BUILD database and Lahey Health system between July 1, 2015 and December 31, 2017 were included in the study. Patients whose stents lacked attributes in AUDI or AccessGUDID associated with their UDIs, ie, drug-eluting status; had invalid medical records; or had multiple stents with mixed drug attributes (both ZES and EES) were excluded from the analysis. Data for this analysis were ingested into the NEST cloud, which is a cloud-based computing environment developed by NESTcc to support active surveillance with RWD. The DELTA 3.7 system was used to perform comparative safety analyses integrating data from both healthcare sources.

The primary outcome was freedom from death at one year. Secondary endpoints included freedom from major adverse cardiovascular events (MACE) at 1 year. MACE is defined as a composite of death, stroke, acute myocardial infarction (AMI), repeat PCI after the index procedure, or the need for coronary artery bypass grafting (CABG). Each component of the composite endpoint was also analyzed separately as secondary endpoints. Prior randomized trials had demonstrated no significant differences in the performance of the two stent types, and there was no suggestion from observational studies or other information sources to suggest that there was a safety concern with either device.^{9–16}

Covariates and outcomes were defined using primarily billing codes from a combination of data sources, and a patient was identified as having a covariate or outcome of interest if any one of the available data sources indicated such.

Lists of International Classification of Diseases 9th and 10th Revision Clinical Modification (ICD-9-CM and ICD-10-CM, respectively) diagnosis codes and Current Procedural Terminology (CPT) codes, termed "value-sets", were created to identify covariates and outcomes of interest (see <u>Tables S1</u> and <u>S2</u>). Each institution then queried their data warehouse to determine if the ICD or CPT codes in the value-sets were added to a patient's medical record in the routine course of care.

Both Mercy and Lahey submit data to the NCDR[®] CathPCI Registry[®] and store the submissions in research databases to supplement RWD collected from electronic health records. In addition to ICD codes, all covariates and some outcomes were partially identified using the original data submitted to NCDR[®]. The specific CathPCI Registry[®] data element sequence numbers referenced for the purposes of this study are provided in <u>Tables S1</u> and <u>S2</u>.

Mortality status was sourced from the EHR at both institutions. The level of missing mortality in the analyzed data was judged to be insignificant in the BUILD study based on similar mortality to that in clinical trials.^{3,9,11} The CERI data warehouse additionally contains death information sourced from the Social Security Death Master File to identify deaths not reported to Lahey nor captured through updates to a patient's medical record.

Statistical Methodology

A propensity score model was constructed using known predictors of post-drug eluting stent (DES) mortality including coronary artery disease (CAD) presentation (categorical), left main (LM) coronary artery stenosis (continuous), source of admission (categorical), history of heart failure (categorical), smoking (categorical), dyslipidemia (categorical), hypertension (categorical), and dialysis (categorical). A complete list of covariates is included in Table 1.

Inverse probability treatment weighting (IPTW) was performed by assigning a weight to each case, derived as the inverse of the probability of receiving the respective treatment, which shrinks the influence of those patients receiving the device of interest proportional to their degree of dissimilarity to those patients receiving the alternative device. Thus, patients who are most similar, as measured by the propensity score, to the patients receiving the alternate device have

99

Baseline Covariate	Ве	fore Weighting	After Weighting					
	EES (n=3334)	ZES (n=1002)	Std Diff	EES (n=3266.6)	ZES (n=921.9)	Std Diff		
Age (Mean ± SD)	65.87 (11.54)	66.87 (11.91)	0.086	66.08 (11.48)	66.32 (11.83)	0.021		
Female	31.10	27.25	0.085	30.04	27.57	0.054		
Hypertension	79.93	83.63	0.096	80.96	83.36	0.061		
Dyslipidemia	78.13	80.64	0.062	78.94	81.54	0.063		
Dialysis	1.86	1.90	0.003	1.84	1.65	0.014		
Diabetes	35.96	33.53	0.051	35.49	35.47	0.001		
Prior MI	41.78	65.97	0.500	47.51	53.95	0.129		
Prior HF	17.64	20.86	0.082	18.40	19.48	0.028		
Smoker	26.15	22.16	0.094	25.17	23.04	0.049		
Emergency	53.33	41.62	0.236	50.34	44.02	0.126		
Stable Angina	6.78	6.69	0.004	6.87	7.92	0.042		
Unstable Angina	36.50	36.73	0.005	36.81	39.15	0.049		
NSTEMI	31.70	28.84	0.062	30.79	27.76	0.066		
STEMI	20.85	22.75	0.046	21.11	19.88	0.030		
LM Stenosis (Mean ± SD)	9.37 (21.43)	8.34 (20.57)	0.049	9.14 (21.13)	9.00 (21.45)	0.007		

Table I Covariate Balance Before and After Weighting

Abbreviations: EES, everolimus-eluting stent; ZES, zotarolimus-eluting stent; Std Diff, absolute standard difference; SD, standard deviation; MI, myocardial infarction; HF, heart failure; NSTEMI, non-ST segment elevation MI; STEMI, ST segment MI; LM, left main coronary artery.

a greater influence on the results, as compared to those patients who are less similar. Weights were trimmed at the 5% and 95% levels to reduce the influence of extreme outlier propensity values. Baseline characteristics after propensity score weighting are described, and standardized differences were calculated as the difference in means or proportions divided by the standard error (SE), with an absolute value <0.10 indicating adequate balance in the distribution of the covariate. For IPTW, the measure of interest was the adjusted odds ratio (OR_{adj}) for the exposure covariate, namely: treatment with EES, for each outcome: mortality and MACE at 12 months. The result was considered significant for increased risk of an adverse event if the OR_{adj} was greater than 1.0, with a 95% CI excluding 1.0.

As a further sensitivity analysis, the data were re-analyzed using propensity score matching survival methods¹⁷ in addition to the IPTW approach. In the propensity-matched analysis, those patients treated with the device of interest (EES) were identified and matched in a 1:1 manner using a greedy matching algorithm with patients treated with the comparator stent (ZES) and using a caliper width equal to 0.2 of the standard deviation of the logit of the propensity score. A matching radius of 90 days (ie, control cases performed up to 90 days prior to, or after, the procedure of interest) was used for this analysis. We used the Kaplan–Meier estimator¹⁹ to estimate the survival curves for freedom from death and MACE events in each of the matched treated and control groups. Time to event was defined as the number of months from the date of treatment with EES in the treated matched group and ZES in the control matched group. We used the Log Rank test²⁰ to compare the survival curves between matched groups for both events. A significance level of <0.05 was used to reject the null hypothesis that there is no difference between the curves.

All analyses were performed within DELTA version 3.7 using R (version 4.0)¹⁸ based packages for propensity weighting, matching, and survival analyses.

Missing covariate data were handled using univariate imputation rules, assuming absence of a condition for dichotomous variables, and using the gender-specific median value for continuous variables. Any covariates with more than 5% missing values were excluded from the propensity score model.

This study involves a secondary analysis of Mercy's BUILD data. BUILD was granted exempt status by the Mercy IRB (ID number 865972-3). The project was submitted to the Lahey IRB for review and was determined not to be human subjects research (ID number 1692573-1).

Results

A total of 4681 patients (Mercy 2637, Lahey 1769) underwent PCI with EES or ZES implantation during the observation period. After applying exclusions, there remained 4336 patients (Mercy 2581, Lahey 1755). A total of 56 Mercy patients were excluded either for invalid records or because they had a second procedure within 14 days of the index procedure. Within the Lahey patients, 14 were excluded as they received both ZES and EES devices during the index procedure. Of the included patients, 3334 patients (76.9%) received EES (Mercy 2356, Lahey 978) and 1002 (23.1%) received ZES (Mercy 225, Lahey 777).

Propensity Weighting

Baseline characteristics of significance and their distribution in patients receiving EES and ZES including age, gender, CAD presentation [stable angina, unstable angina, non-ST elevation myocardial infarction (NSTEMI), or STEMI], most severe percent stenosis in the LM coronary artery, source of admission, history of myocardial infarction, heart failure, diabetes mellitus, smoking history, dyslipidemia, hypertension, and current dialysis are shown in Table 1. The treated populations were similar between the two organizations, and the overall covariate distributions are typical of contemporary interventional cardiology practice, with a mean age of approximately 66 years, 70% male and 34% with a history of diabetes at the time of their index procedure. The results of propensity score weighting are also shown in Table 1.

Prior to weighting, there were significant imbalances between the stent treatment groups in the proportion of patients with "prior myocardial infarction" and "emergency procedures". After IPTW, residual standardized differences were significantly improved, with only 2 of 15 covariates having residual imbalance, defined as a standardized mean difference of >0.1. These two covariates, which are both associated with higher risk of mortality and MACE, were imbalanced in opposite directions, with a higher proportion of ZES treated patients having a prior history of myocardial infarction, and a larger proportion of EES treated patients undergoing the stent procedure as an emergency.

Mortality

One-year mortality was not significantly different in patients receiving EES compared to those receiving ZES [adjusted odds ratio = 0.97 (95% CI 0.81-1.17); p = 0.780].

MACE

The one-year incidence of MACE was not significantly different in patients receiving EES compared to those receiving ZES [adjusted odds ratio = 1.04 (95% CI 0.92-1.16; p = 0.551)].

Propensity Matching

Table 2 shows the results of the propensity score matching analysis.

Matches to patients receiving EES were found for 992 of the 1002 patients receiving EES, with absolute standardized differences ranging from 0.002 to 0.077 for the 15 covariates (well less than the pre-specified 0.10 criterion). Unmatched EES patients were more likely than ZES patients to be female and to have had an emergency procedure.

Mortality

One-year survival curves for matched patients receiving only EES ("treated") and those receiving only ZES ("control") along with 95% confidence bands (CB) are shown in Figure 1. The CB overlap at all time points, indicating there is no significant difference in one-year mortality in patients receiving EES (547 of 992 of EES patients remained available for analysis, after censoring, at the end of 12 months) compared to ZES (546 of 992 patients). The Log-Rank statistic was 0.3348 (p = 0.563) indicating no difference in survival curves.

MACE

One-year MACE-free survival curves for matched patients receiving only EES ("treated") and those receiving only ZES ("control") along with 95% confidence bands (CB) are shown in Figure 2. The CB overlap at all time points, indicating

Baseline	В	efore Match		4	After Match	Unmatched		
Characteristics	EES (n=3334)	ZES (n=1002)	Std Diff	EES (n=992)	ZES (n=992)	Std Diff	EES (n=2342)	Std Diff
Age (Mean ± SD)	65.87 (11.54)	66.87 (11.91)	0.086	66.48 (11.19)	66.77 (11.88)	0.025	65.6 (11.68)	0.077
Female (%)	1037 (31.1)	273 (27.25)	0.085	241 (24.29)	272 (27.42)	0.071	796 (33.99)	0.215
Hypertension (%)	2665 (79.93)	838 (83.63)	0.096	821 (82.76)	829 (83.57)	0.022	1844 (78.74)	0.102
Dyslipidemia (%)	2605 (78.13)	808 (80.64)	0.062	812 (81.85)	801 (80.75)	0.028	1793 (76.56)	0.131
Diabetes (%)	1199 (35.96)	336 (33.53)	0.051	343 (34.58)	336 (33.87)	0.015	856 (36.55)	0.041
Dialysis (%)	62 (1.86)	19 (1.9)	0.003	17 (1.71)	19 (1.92)	0.015	45 (1.92)	0.016
Prior MI (%)	1393 (41.78)	661 (65.97)	0.500	669 (67.44)	651 (65.62)	0.039	724 (30.91)	0.785
Prior HF (%)	588 (17.64)	209 (20.86)	0.082	190 (19.15)	206 (20.77)	0.040	398 (16.99)	0.056
Smoker (%)	872 (26.15)	222 (22.16)	0.094	219 (22.08)	222 (22.38)	0.007	653 (27.88)	0.134
Emergency (%)	1778 (53.33)	417 (41.62)	0.236	455 (45.87)	417 (42.04)	0.077	1323 (56.49)	0.214
Stable Angina (%)	226 (6.78)	67 (6.69)	0.004	68 (6.85)	67 (6.75)	0.004	158 (6.75)	0.004
Unstable Angina (%)	1217 (36.5)	368 (36.73)	0.005	365 (36.79)	364 (36.69)	0.002	852 (36.38)	0.009
NSTEMI (%)	1057 (31.7)	289 (28.84)	0.062	299 (30.14)	289 (29.13)	0.022	758 (32.37)	0.048
STEMI (%)	695 (20.85)	228 (22.75)	0.046	213 (21.47)	222 (22.38)	0.022	482 (20.58)	0.022
LM Stenosis (Mean ± SD)	9.37 (21.43)	8.34 (20.57)	0.049	8.85 (20.89)	8.41 (20.65)	0.021	9.59 (21.65)	0.035

|--|

Abbreviations: EES, everolimus-eluting stent; ZES, zotarolimus-eluting stent; Std Diff, absolute standard difference; SD, standard deviation; MI, myocardial infarction; HF, heart failure; NSTEMI, non-ST segment elevation MI; STEMI, ST segment MI; LM, left main coronary artery.

that there is no significant difference in one-year MACE in patients receiving EES (484 of 992 EES patients were free from MACE and remained available for analysis at the end of 12 months) compared to ZES (510 of 992 patients). The Log-Rank statistic was 0.0534 (p = 0.817) indicating no difference in survival curves.



NUMBER AT RISK														
Durations	0	1	2	з	4	5	6	7	8	9	10	11	12	13
Treated	992	992	934	918	909	901	896	893	889	884	878	874	867	547
Control	992	992	927	920	909	901	893	888	881	873	867	867	865	546

Figure I Survival Summary Report: Freedom from Death at 12-months - EES vs ZES.



NUMBER AT RISK														
Durations	0	1	2	з	4	5	6	7	8	9	10	11	12	13
Treated	992	992	915	898	884	872	862	858	850	832	823	803	784	484
Control	992	992	904	892	879	868	857	849	839	826	819	814	807	510

Figure 2 Survival Summary Report: Freedom from Major Adverse Cardiac Events (MACE) at 12-months - EES vs ZES.

Conclusion and Discussion

In this retrospective, observational, analysis, we found low rates of adverse events and no difference in the one-year safety profiles of the predominant drug-eluting stents available in the United States during the study period. The current study demonstrates the feasibility of leveraging cloud-based active surveillance of multi-health system real-world data sources in the evaluation of comparative safety of commonly used cardiovascular devices. Reassuringly, no meaningful differences in the 1-year rates of death, MACE or the individual components of MACE between ZES and EES were identified using either the primary IPTW analysis or using the propensity-matched survival methods. These results are consistent with those of randomized controlled trials^{9–16} comparing the two stent types as well as the findings of Mercy's prior analysis.³

These findings also demonstrate the potential value of applying active surveillance tools, such as DELTA, to real-world EHR data from multiple healthcare systems within a cloud computing environment. Routinely applying active surveillance tools to EHR data may provide physicians, patients, regulators, and manufacturers with near real-time assessment of medical device performance. EHR-based surveillance, though challenged by data quality and completeness issues, offers unique advantages in terms of granularity as well as timeliness, relative to other commonly used real-world data sources, such as claims data and registry data. The results of the current demonstration support the continued development of cloud-based, multiple health system medical device surveillance systems as part of enhanced post-market medical device evaluation efforts.

Limitations

As with all observational studies, the risk of residual confounding may have influenced the results of this analysis. Propensity scoring and matching methods were used to mitigate this possibility, though minor residual imbalances in prior myocardial

infarction and emergency remained that may have impacted the results. Separately, this analysis was performed on retrospective data simulating a prospective study, thereby introducing an additional potential bias related to more complete caselevel information being available at the time the analysis was performed than would have likely been available if the data had been analyzed with accruing data. Delays in the availability of data for surveillance may be due to refresh rate intervals for data transformation (4–6 weeks at Lahey), as well as scheduled data cleaning and verification steps. Thus, in an optimized prospective active surveillance analysis, it would be expected that incomplete data availability would reduce the power of the analysis earlier in the course of the study, with expected convergence toward the current findings as the RWD were captured for all cases. Additionally, although we included a robust set of baseline covariates in our analysis, some variables that might be expected to affect mortality, eg, the number of implanted stents and adherence to antiplatelet and other drugs, were not captured. There is, however, no reason to expect that these would be significantly different between the two stent types. Also, our primary goal was to identify any potential safety signal, which, if found, would require further evaluation that would include factors not easily obtainable from real-world data sources.

From a technical perspective, the EHR data were ingested into a single cloud environment, then processed and pooled prior to processing by DELTA. The data harmonization enabling analysis from multiple health system sources was a good proxy for the challenges that will be associated with future work that is expected to be more complex when active surveillance is applied to medical devices that are less well characterized and when more complex privacy preserving techniques and analytic architectures are required by participating health systems.

Data Sharing Statement

The patient-level data used in this study contain protected health information and were made available specifically for this study under data use agreements developed for this project between the stakeholder organizations. As such, these data are not available for external analysis without the execution of additional data use agreements, subject to the consent of the data owner organizations to participate in the additional data use agreement.

Acknowledgments

Kim Collison-Farr coordinated the Mercy team effort and assisted in Mercy data identification and ascertainment. Catherine Axtell conducted much of the original Mercy SAS code validation.

Funding

This work was supported, in part, by the Food and Drug Administration through a grant (1 U01 FD 006292-01). The views expressed do not necessarily reflect the official policies of the Department of Health and Human Services; nor does any mention of trade names, commercial practices, or organization imply endorsement by the United States Government.

Disclosure

Dr Resnic reports having received grants from the NIH and the William Wood Foundation. The other authors have nothing to declare for this work.

References

- 1. U.S. Food and Drug Administration. Medical device safety action plan: protecting patients, promoting public health. fda.gov. Available from: https:// www.fda.gov/files/about%20fda/published/Medical-Device-Safety-Action-Plan-Protecting-Patients-Promoting-Public-Health-%28PDF%29.pdf. Accessed February 13, 2024.
- National Evaluation System for health Technologies. Developing the NESTcc active surveillance portfolio. NESTcc; 2018. Available from: https:// nestcc.org/developing-active-surveillance/. Accessed February 13, 2024.
- Drozda JP, Graham J, Muhlestein JB, et al. Multi-institutional distributed data networks for real-world evidence about medical devices: building unique device identifiers into longitudinal data (BUILD). JAMIA Open. 2022;2:00ac035. Available at Multi-institutional distributed data networks for real-world evidence about medical devices: building unique device identifiers into longitudinal data (BUILD) | JAMIA Open | Oxford Academic (oup.com). doi:10.1093/jamiaopen/o0ac035
- 4. Drozda JP, Roach J, Forsyth T, et al. Constructing the informatics and information technology foundations of a medical device evaluation system: a report from the FDA unique device identifier demonstration. J Am Med Inform Assoc. 2018;25(2):111–120. doi:10.1093/jamia/ocx041
- Platt R, Brown JS, Robb M, et al. The FDA Sentinel Initiative an evolving national resource. N Engl J Med. 2018;379(22):2091–2093. doi:10.1056/ NEJMp1809643

- National Cardiovascular Data Registry. CathPCI Registry[®] coders data dictionary. Available from: https://www.ncdr.com/WebNCDR/docs/defaultsource/public-data-collection-documents/cathpci v4 codersdictionary 4-4.pdf?sfvrsn=b84d368e 2. Accessed January 19, 2022.
- 7. United States Food and Drug Administration, AccessGUDID. Available from: https://accessgudid.nlm.nih.gov/. Accessed January 13, 2022.
- Tcheng JE, Crowley J, Tomes M, et al. Unique device identifiers (udis) for coronary stent post-market surveillance and research: a report from the FDA's Medical Device Epidemiology Network (MDEpiNet) udi demonstration. Am Heart J. 2014;168(4):405–13, e2. doi:10.1016/j.ahj.2014.07.001
- 9. Serruys PW, Silber S, Garg S, et al. Comparison of zotarolimus-eluting and everolimus-eluting coronary stents. N Engl J Med. 2010;363 (2):136-146. doi:10.1056/NEJMoa1004130
- 10. Iqbal J, Serruys P, Silber S, et al. Comparison of zotarolimus- and everolimus-eluting coronary stents: final 5-year report of the RESOLUTE all-comers trial. *Circ Cardiovasc Interv.* 2015;8(6):e002230. doi:10.1161/CIRCINTERVENTIONS.114.002230
- 11. von Birgelen C, Basalus MW, Tandjung K, et al. A randomized controlled trial in second-generation zotarolimus eluting Resolute stents versus everolimus eluting Xience V stents in real-world patients: the TWENTE trial. J Am Coll Cardiol. 2012;59(15):1350–1361. doi:10.1016/j.jacc.2012.01.008
- 12. Tandjung K, Sen H, Lam M, et al. Clinical outcome following stringent discontinuation of dual antiplatelet therapy after 12 months in real-world patients treated with second generation zotarolimus-eluting Resolute and everolimus-eEluting Xience V stents: 2-year follow-up of the randomized TWENTE trial. J Am Coll Cardiol. 2013;61(24):2406–2416. doi:10.1016/j.jacc.2013.04.005
- von Birgelen C, Sen H, Lam MK, et al. Third-generation zotarolimus- eluting and everolimus-eluting stents in all-comer patients requiring a percutaneous coronary intervention (Dutch PEERS): a randomised, single-blind, multicentre, non-inferiority trial. *Lancet.* 2014;383(9915):413–423. doi:10.1016/ S0140-6736(13)62037-1
- 14. Park KW, Kang S-H, Kang H-J, et al. HOST–ASSURE Investigators. A randomized comparison of platinum chromium-based everolimus-eluting stents versus cobalt chromium based Zotarolimus-Eluting stents in all-comers receiving percutaneous coronary intervention: HOST-ASSURE (harmonizing optimal strategy for treatment of coronary artery stenosis-safety & effectiveness of drug-eluting stents & anti-platelet regimen), a randomized, controlled, noninferiority trial. J Am Coll Cardiol. 2014;63(25 Pt A):2805–2816. doi:10.1016/j.jacc.2014.04.013
- 15. Chen KY, Rha SW, Wang L, et al. Unrestricted use of 2 new-generation drug-eluting stents in patients with acute myocardial infarction: a propensity score-matched analysis. J Am Coll Cardiol Intventions. 2012;9(9):936–945. doi:10.1016/j.jcin.2012.05.009
- Park K, Lee J, Kang S-H, et al. Everolimus-eluting Xience V stents versus zotarolimus-eluting Resolute stents in real-world practice patient-related and stent-related outcomes from the multicenter prospective EXCELLENT and RESOLUTE-Korea registries. J Am Coll Cardiol. 2013;61(5):536–544. doi:10.1016/j.jacc.2012.11
- 17. Resnic FS, Majithia A, Dhruva SS, et al. Active Surveillance of the ICD Registry for Defibrillator Lead Failures. *Circ Cardiovasc Qual Outcomes*. 2020;13(4). doi:10.1161/CIRCOUTCOMES.119.006105
- 18. R Core Team. R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria; 2020. Available from: https://www.R-project.org/. Accessed February 13, 2024.
- 19. Goel MK, Khanna P, Kishore J. Understanding survival analysis: Kaplan-Meier estimate. *Int J Ayurveda Res.* 2010;1(4):274–278. PMID: 21455458; PMCID: PMC3059453. doi:10.4103/0974-7788.76794
- 20. Bland JM, Altman DG. The logrank test. BMJ. 2004;328(7447):1073. doi:10.1136/bmj.328.7447.1073

Medical Devices: Evidence and Research

Dovepress

DovePress

Publish your work in this journal

Medical Devices: Evidence and Research is an international, peer-reviewed, open access journal that focuses on the evidence, technology, research, and expert opinion supporting the use and application of medical devices in the diagnosis, monitoring, treatment and management of clinical conditions and physiological processes. The identification of novel devices and optimal use of existing devices which will lead to improved clinical outcomes and more effective patient management and safety is a key feature of the journal. The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/medical-devices-evidence-and-research-journal

fl 🔰 in 🗖